\begin{table}%t5
\caption{Millimetre measurements and IRAS photometry compiled from the literature. The columns~\beam\ list 
the {\it HPBW}; note that the IRAS PSF sizes are given in arcmin.}
\label{fluxliterature} 
%\centerline
{
\begin{tabular}{ l | llcc | llcc | llcc | llcc | llcc }
\hline\hline
$\;\;\;~\lambda$ & 
\multicolumn{1}{|c}{Flux}  & 
\multicolumn{1}{c}{Error} & 
\multicolumn{1}{c}{\beam} & 
Ref. &
\multicolumn{1}{|c}{Flux}  & 
\multicolumn{1}{c}{Error} & 
\multicolumn{1}{c}{\beam} & 
Ref. & 
\multicolumn{1}{|c}{|Flux}  & 
\multicolumn{1}{c}{Error} & 
\multicolumn{1}{c}{\beam} & 
Ref. & 
\multicolumn{1}{|c}{Flux}  & 
\multicolumn{1}{c}{Error} & 
\multicolumn{1}{c}{\beam} & 
Ref. & 
\multicolumn{1}{|c}{Flux}  & 
\multicolumn{1}{c}{Error} & 
\multicolumn{1}{c}{\beam} & 
Ref. \\
(mm)  & 
\multicolumn{1}{|c}{(Jy)}     & 
\multicolumn{1}{c}{(Jy)}     & 
\multicolumn{1}{c}{(\arcsec)}&   
&
\multicolumn{1}{|c}{(Jy)}     &
\multicolumn{1}{c}{(Jy)}     & 
\multicolumn{1}{c}{(\arcsec)}&       
& 
\multicolumn{1}{|c}{(Jy)}     &
\multicolumn{1}{c}{(Jy)}     & 
\multicolumn{1}{c}{(\arcsec)}&      
&
\multicolumn{1}{|c}{(Jy)}     &
\multicolumn{1}{c}{(Jy)}     & 
\multicolumn{1}{c}{(\arcsec)}&      
&
\multicolumn{1}{|c}{(Jy)}     & 
\multicolumn{1}{c}{(Jy)}     &
\multicolumn{1}{c}{(\arcsec)}&
\\\hline 
% first entry line
&\multicolumn{4}{|c|}{\bf{RY Tau}}  & 
 \multicolumn{4}{|c}{\bf{FT Tau}}  & 
 \multicolumn{4}{|c}{\bf{DG Tau B}}& 
 \multicolumn{4}{|c}{\bf{DG Tau}}  & 
 \multicolumn{4}{|c}{\bf{HL Tau}}   \\
0.012 & 17.74& 0.027& 1\am& (a)  & 0.36 & 0.045& 1\am& (a)  &      &      &     &      &      &      &     &      &  9.74& 0.031& 1\am& (a) \\
0.025 & 26.48& 0.054& 1\am& (a)  & 0.57 & 0.037& 1\am& (a)  &      &      &     &      & 30.39& 0.065& 1\am& (a)  & 31.18& 0.073& 1\am& (a) \\
0.060 & 18.91& 0.066& 2\am& (a)  & 0.82 & 0.041& 2\am& (a)  &      &      &     &      & 31.28& 0.039& 2\am& (a)  & 76.26& 0.810& 2\am& (a) \\
0.100 & 13.50& 2.501& 4\am& (a)  &      &      &     &      &      &      &     &      & 39.86& 0.840& 4\am& (a)  & 77.95& 1.394& 4\am& (a) \\
1.056 & 0.354& 0.036& 22  & (b)  &      &      &     &      &      &      &     &      & 0.489& 0.033& 22  & (b)  &  1.13& 0.02 & 22  & (b) \\
1.1   & 0.28 & 0.09 & 18.5& (f)  &      &      &     &      &      &      &     &      & 0.57 & 0.07 & 18.5& (f)  &  1.11& 0.02 & 18  & (k) \\
1.2   & 0.212& 0.021& 12  & (g)  &      &      &     &      &      &      &     &      & 0.420& 0.042& 12  & (g)  & 0.961& 0.096& 12  & (g) \\
1.3   & 0.229& 0.017& 11  & (c)  & 0.130& 0.014& 11  & (c)  &      &      &     &      & 0.443& 0.020& 11  & (c)  & 0.879& 0.019& 11  & (c) \\
2.0   & 0.052& 0.006&  5  & (d)  &      &      &     &      &      &      &     &      & 0.14 & 0.03 & 27.5& (f)  & 0.161& 0.017& 5   & (d) \\
2.6   &      &      &     &      &      &      &     &      &0.0691&0.0017&  4  & (h)  &      &      &     &      &      &      &     &     \\
2.7   &      &      &     &      & 0.025&0.0022&  3  & (e)  &0.0494&0.0088& 0.7 & (i)  & 0.059&0.0020& 3   & (e)  &0.1069&0.0078& 0.6 & (i) \\
3.0   &      &      &     &      &      &      &     &      &      &      &     &      & 0.057&0.0040& 6   & (j)  &0.074 &0.0040& 3   & (j) \\
\hline
&\multicolumn{4}{|c}{\bf{UZ Tau E}} &
 \multicolumn{4}{|c}{\bf{DL Tau}}   &
 \multicolumn{4}{|c}{\bf{CI Tau}}   &
 \multicolumn{4}{|c}{\bf{DO Tau}}   &
 \multicolumn{4}{|c}{\bf{GM Aur}}   \\
0.012 &  1.38& 0.031& 1\am& (a)  & 0.97 & 0.034& 1\am& (a)  & 0.78 & 0.025& 1\am& (a)  &      &      &     &      & 0.25 & 0.031& 1\am& (a) \\
0.025 &  1.76& 0.043& 1\am& (a)  & 1.32 & 0.045& 1\am& (a)  & 1.30 & 0.046& 1\am& (a)  & 4.07 & 0.041& 1\am& (a)  & 1.07 & 0.042& 1\am& (a) \\
0.060 &  2.37& 0.069& 2\am& (a)  & 1.39 & 0.084& 2\am& (a)  & 2.15 & 0.071& 2\am& (a)  & 6.33 & 0.106& 2\am& (a)  & 3.08 & 0.112& 2\am& (a) \\
0.100 &      &      &     &      &      &      &     &      &      &      &     &      & 8.55 & 1.483& 4\am& (a)  &      &      &     &     \\  
1.056 & 0.213& 0.039& 22  & (b)  & 0.273& 0.037& 22  & (b)  & 0.217& 0.050& 22  & (b)  & 0.194& 0.032& 22  & (b)  &      &      &     &     \\
1.1   &      &      &     &      & 0.26 & 0.03 & 18.5& (f)  &      &      &     &      & 0.18 & 0.02 & 18.5& (f)  & 0.38 & 0.03 & 18  & (k) \\
1.2   & 0.170& 0.017& 12  & (g)  & 0.198& 0.020& 12  & (g)  & 0.177& 0.018& 12  & (g)  & 0.145& 0.015& 12  & (g)  &      &      &     &     \\
1.3   & 0.137& 0.028& 1.1 & (l)  & 0.230& 0.014& 11  & (c)  & 0.190& 0.017& 11  & (c)  &0.1375& 0.032&  3  & (m)  & 0.253& 0.012& 11  & (c) \\
1.35  &      &      &     &      &      &      &     &      &      &      &     &      &0.0986& 0.024&  4  & (m)  &      &      &     &     \\
2.0   &      &      &     &      & 0.12 & 0.03 & 27.5& (f)  &      &      &     &      & 0.037& 0.005&  5  & (d)  & 0.037& 0.004& 5   & (d) \\
2.7   &0.0255&0.0016& 3   & (e)  &0.0296&0.0020&  3  & (e)  &0.0310&0.0033& 3   & (e)  &0.0255&0.0027&  3  & (e)  &0.0277&0.0022& 3   & (e) \\
3.0   &      &      &     &      & 0.023&0.0040&  8  & (j)  &      &      &     &      &      &      &     &      &      &      &     &     \\
3.1   & 0.014&0.0031& 2.1 & (l)  &      &      &     &      &      &      &     &      &      &      &     &      &      &      &     &     \\
3.4   &      &      &     &      &      &      &     &      &      &      &     &      &0.0142&0.0007&  1  & (m)  &      &      &     &     \\
\hline
\end{tabular}}
%
 References --- 
        (a)~\citet{weaver92};
        (b)~\citet{beckwith91};
	(c)~\citet{beckwith90};
	(d)~\citet{kitamura02};
	(e)~\citet{dutrey96};
	(f)~\citet{mannings94};
	(g)~\citet{altenhoff94};
	(h)~\citet{mitchell97};
	(i)~\citet{looney00};
	(j)~\citet{ohashi96};
	(k)~\citet{adams90};
	(l)~\citet{jensen96};
	(m)~\citet{koerner95}.
%
\end{table}